**Author's response to reviews**

**Title:** Novel PTRF-CAVIN mutation in a child with mild myopathy and very mild congenital lipodystrophy

**Authors:**

- Anna Ardissone (anna.ardissone@istituto-besta.it)
- Cinzia Bragato (cinzia.bragato@istituto-besta.it)
- Lorella Caffi (npibergamo@ospedaliriuniti.bergamo.it)
- Flavia Blasevich (flavia.blasevich@istituto-besta.it)
- Lucia Morandi (lmorandi@istituto-besta.it)
- Isabella Moroni (isabella.moroni@istituto-besta.it)
- Marina Mora (mmora@istituto-besta.it)

**Version:** 4 **Date:** 11 January 2013

**Author's response to reviews:** see over
To: BMC Medical Genetics

RE: Manuscript MS: 1050092838846767 - Novel PTRF-CAVIN mutation in a child with mild myopathy and very mild congenital lipodystrophy.

Dear Editor,

We have made the following changes as requested:

In the “Consent” section the following sentence has been added: "Parental written informed consent was obtained for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor of this journal."

We are uploading the corrected version of the manuscript.

Sincerely yours,

Marina Mora, PhD
Dept. of Neuromuscular Diseases
Istituto Nazionale Neurologico “C. Besta”
Via Temolo 4
20126 Milano
Italy
Tel: +39 02 23942632
fax: +39 02 23942619
Email: mmora@istituto-besta.it